Author's response to reviews

Title: Social Networks, work and Network-Based Resources for the Management of Long Term Conditions: a framework and study protocol for developing self care support

Authors:

Anne Rogers (anne.rogers@manchester.ac.uk)
Ivaylo Vassilev (ivaylo.vassilev@manchester.ac.uk)
Caroline Sanders (caroline.sanders@manchester.ac.uk)
Susan Kirk (susan.kirk@manchester.ac.uk)
Carolyn Chew-Graham (carolyn.chew-graham@manchester.ac.uk)
Anne Kennedy (anne.kennedy@manchester.ac.uk)
Joanne Protheroe (joanne.protheroe@manchester.ac.uk)
Peter Bower (peter.bower@manchester.ac.uk)
Christian Blickem (christian.blickem@manchester.ac.uk)
David Reeves (david.reeves@manchester.ac.uk)
Dharmi Kapadia (dharmi.kapadia@manchester.ac.uk)
Helen Brooks (helen.brooks@manchester.ac.uk)
Catherine Fullwood (catherine.fullwood@manchester.ac.uk)
Gerry Richardson (gar2@york.ac.uk)

Version: 10 Date: 24 May 2011

Author's response to reviews:

Abstract: This should be identical in the manuscript and the submission system

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. Currently Catherine Fullwood’s name is spelt Catherine Fulwood in the submission system.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.